Author's response to reviews

Title: Hyperoxia retards growth and induces apoptosis and loss of glands and blood vessels in DMBA-induced rat mammary tumors

Authors:

Anette Raa (anette.raa@biomed.uib.no)
Christine Stansberg (christine.stansberg@helse-bergen.no)
Vidar M Steen (vidar.martin.steen@helse-bergen.no)
Rolf Bjerkvig (rolf.bjerkvig@biomed.uib.no)
Rolf K Reed (rolf.reed@biomed.uib.no)
Linda EB Stuhr (linda.stuhr@biomed.uib.no)

Version: 4 Date: 29 January 2007

Author's response to reviews:

I have made the following formatting changes:
1. Removed full stop from the end of the title.
2. Removed websites from the text.
3. Removed et al in the reference list and included all authors.
4. Tables are now in new numerical order due to a new table 1 as asked for.
5. Figures are cropped to the best of my ability.